Author's response to reviews

Title: Ventilatory chemoreflexes and the apnea-hypopnea index in six-to-twelve year old children

Authors:

Dr Ralph F Fregosi Ph.D. (Fregosi@u.arizona.edu)
Dr Stuart F Quan M.D. (squan@resp-sci.arizona.edu)
Andrew C Jackson Ph.D. (ajax@bu.edu)
Kris L Kaemingk Ph.D. (kaemingk@peds.arizona.edu)
Wayne J Morgan M.D. (wmorgan@resp-sci.arizona.edu)
Dr Jamie L Goodwin Ph.D. (jamieg@resp-sci.arizona.edu)
Jenny C Reeder B.S. (reederjc@yahoo.com)
Rosaria K Cabrera B.A. (rmkelley@email.arizona.edu)
Elena Antonio (eantonio@u.arizona.edu)

Version: 6 Date: 28 Apr 2004

All changes have been made. Please note that I am using EndNote 6.0.2 for the references, and am formatting using the Biomed central link; the result is as you see it. I had the same problem with BMC physiology, and they were supposed to look into this issue. I will not format the references by hand, so please either correct the problem on your end, or tell me another compatible format within the EndNote program.

Sincerely,

Ralph Fregosi